Credit Constraints, Firms' Precautionary Investment, and the Business Cycle^{*}

Ander Perez^{\dagger}

Universitat Pompeu Fabra and Barcelona GSE

November 20, 2012

Abstract

This paper studies the macroeconomic implications of firms' investment composition choices in the presence of credit constraints. Following a negative and persistent aggregate productivity shock, firms shift into short-term investments because they produce more pledgeable output and because they help alleviate future borrowing constraints. This produces a short-run dampening of the effects of the shock, at the expense of lower long-term investment and future output, relative to an economy with no credit market imperfections. The effects are exacerbated by a steepening of the term structure of interest rates that further encourages a shift towards short-term investments in the short-run. Small temporary shocks to the severity of financing frictions generate large and long-lasting effects on output through their impact on the composition of investment. A positive financial shock produces much stronger effects than an identical negative shock, while the responses to positive and negative shocks to aggregate productivity are roughly symmetric. Finally, the paper introduces a novel explanation for the countercyclicality of financing constraints of firms.

Keywords: Investment composition; Financial frictions; Business cycles; Idiosyncratic production risk; Firm heterogeneity

JEL Classification Numbers: D92, E22, E32, E44, G31, G32

^{*}I am grateful to Kosuke Aoki and Nobu Kiyotaki for very valuable discussions, and for very helpful comments Vicente Cunat, Wouter den Haan, Antoine Faure-Grimaud, Roman Inderst, Eric Leeper, Albert Marcet, Raoul Minetti, Alex Michaelides, Javier Suarez, Aleh Tsyvinski, David C. Webb, Alwyn Young and seminar participants in LSE, HEC Montreal, Indiana, CEMFI, Bank of Spain, Alicante, Carlos III, Autonoma de Barcelona, IAE-CSIC, Tilburg, Cambridge, Birkbeck, Pompeu Fabra, Bank of England, the European Finance Association Meetings 2009 (Bergen), the II Bank of Italy Conference on Macro Modelling, and the Society for Economic Dynamics Meetings 2010 (Montreal). A previous version of this paper circulated under the title "Firms' Self-Insurance and the Financial Accelerator".

[†]Email address: ander.perez@upf.edu

1 Introduction

How do financial constraints affect the dynamics of the aggregate composition of investment? Does the behavior of the composition of investment in the presence of financing constraints have important consequences for how such frictions influence the response of an economy to shocks? Firm managers typically face a choice between multiple projects that may differ in terms of their duration, liquidity risk and pledgeability of returns, amongst other characteristics, and financing constraints may have important implications for firms' investment composition choices along these dimensions. Indeed firm-level empirical work has found that credit constrained firms tend to invest in safer and shorterterm projects¹, display a more procyclical pattern of their share of R&D investment², and purchase used, rather than new, capital³. At the aggregate level, Dew-Becker (2011) shows that the average duration of aggregate investment falls in downturns, and Aghion et al. (2010) provide evidence that long-term investment as a share of total investment decreases following shocks that can be expected to make firms more credit constrained, and also document that this effect is stronger for less financially developed economies.

In this paper we perform a quantitative theoretical exploration of the implications for aggregate investment and output dynamics of the interaction between investment composition choice, idiosyncratic risk and financing constraints. More specifically, the dimension of project choice that will be studied is the duration or maturity of investment projects, which in turn generates endogenous and time-varying implications for other project characteristics such as liquidity risk and pledgeability of returns. We first develop a model which produces results consistent with the evidence described above and use it to address several questions. Does the interaction between investment choice, idiosyncratic risk and financing constraints dampen or amplify the short-run effects of aggregate productivity shocks? Does it influence how the effects of productivity shocks propagate through time? How does it influence the impact of financial shocks?

These questions are dealt with by analyzing a dynamic general equilibrium model of an economy in which heterogeneous firms produce a consumption good using a short-term and a long-term technology and labor provided by a representative household. Firms suffer idiosyncratic liquidity shocks proportional to their total capital stock and face an endogenous constraint that limits the amount of debt finance they can receive to a fraction of the present value of their lowest possible realization of future earnings. The nature of the liquidity shock means that long-term investment increases future liquidity risk more than short-term investment, and the borrowing constraint implies that longterm investment typically produces less pledgeable output on the margin than short-term

¹See Chevalier and Scharfstein (1996), Andrade and Kaplan (1998), Peyer and Shivdasani (2001), and Ahn et al. (2006).

²See Aghion et al. (2012).

³See Eisfeldt and Rampini (2007).

investment.

Firms have an incentive to shift investment towards the long-term project in recessions because its returns are relatively acyclical whereas short term activities offer poor returns in downturns. In the presence of financing constraints, this decreases the amount of pledgeable output for a given level of total investment and renders the severity of these constraints countercyclical. This explanation for the countercyclicality of financing constraints is the first result of the paper and provides an alternative to the two main explanations offered in the literature, based on countercyclical agency costs on one hand⁴, and collateral constraints and lack of indexation of debt contracts on the other⁵.

The second contribution of the paper is the description and quantitative evaluation of a novel mechanism that delivers short-run dampening of productivity shocks and longrun propagation. If a negative unexpected aggregate productivity shock hits the economy, credit constraints tighten for firms and this causes constrained firms to shift into short-term investment because it produces relatively more pledgeable output.⁶ Credit constraints are expected to remain tight for some time and this anticipation of future constraints further encourages constrained firms to shift into short-term investment because long-term investment carries more liquidity risk and increases the probability of suffering future credit constraints.⁷ The interaction of investment duration choice and financing constraints acts as a financial decelerator that dampens the contemporaneous response to the negative productivity shock.

The behavior of the interest rate exacerbates these effects. In the analysis we compare the transitional dynamics following the negative temporary shock of an economy where the interest rate is assumed fixed with one in which it is endogenous. Interest rates initially drop moderately following the shock for very few periods, and subsequently rise sharply, so the term structure of interest rates steepens considerably following the shock. This makes long-term investment relatively less attractive than short-term investment following the shock, and is an additional force that encourages a further shift out of longterm investment in the short run. While the behavior of the interest rate is similar in economies with and without credit frictions, their effect on the composition of investment

⁴As in Bernanke and Gertler (1989), Carlstrom and Fuerst (1997), or Christiano et al. (2010).

⁵As in Kiyotaki and Moore (1997) or Iacoviello (2005).

⁶Credit constraints tighten following the negative productivity shock due to the novel mechanism outlined above based on the procyclicality of pledgeable output as a result of the incentive to shift to long-term activities in downturns, but also because of two other more standard reasons. One is the large decrease in firms' wealth following the unexpected shock. Another is the increase in interest rates which decreases the present value of pledgeable output and tightens borrowing constraints.

⁷The importance for firm behavior of the anticipation of future financing constraints has been the focus of recent work. Surveys by Graham and Harvey (2001) and Bancel and Mittoo (2004) find that CFOs consider financial flexibility (having enough internal funds to avoid having to fore-go positive Net Present Value projects in the future) to be the primary determinant of their policy decisions. Almeida, Campello and Weisbach (2004) report that the expectation of future financing problems significantly affects firms' investment policies, and Caggese and Cunat (2007) find that it significantly impacts hiring decisions.

is stronger in the presence of credit constraints because the large rise in interest rates in the medium-term tightens future borrowing constraints and the anticipation of such constraints induces a precautionary reaction of firms in the short-run to further shift out of long-term investment.⁸ In the version of the model with fixed interest rates the maximum output drop following the negative technology shock is 22% smaller in a model featuring financing constraints relative to one in which financial markets are frictionless. When we endogenize the interest rate, the dampening effect rises to 31%.

The decrease in the share of long-term investment propagates the effect of the shock through time. Convex costs of adjustment in investment mean that the large initial fall in the stock of long-term capital takes a long time to recover. The fall is larger with financing constraints, and the recovery is slower because adjustment costs and investment are hindered by the constraints on financing some firms face. As a result, a trade-off arises between contemporaneous amplification and long-term propagation of the effects of shocks; stronger dampening is associated to larger propagation.

A financial shock, in the form of a tightening of financing constraints, generates a large shift out of long-term investment. Even if the financial shock is short-lived, the large drop in the stock of long-term capital takes a long time to recover and propagates the effects of the financial shock for many periods after the severity of financing constraints subsides. Also in this case, general equilibrium effects through the response of the interest rate result in significantly larger effects. Following the shock the slope of the term structure of interest rates increases sharply and further encourages a shift out of long-term investment. A similar contemporaneous dampening effect is at play in the case of a financial shock as well because the shift into short-term investment raises current output at the expense of future output.

The nature of our solution method allows us to study whether the sign of the shock influences the size and shape of the response of the economy to shocks. While the reaction to positive and negative technology shocks is relatively symmetric, the reaction to a tightening and a loosening of credit constraints is highly asymmetric. A loosening of credit constraints generates an increase in output more than three times the size, in absolute value, than the decrease in output following a symmetric tightening of constraints. The intuition for this large asymmetry is that in equilibrium firms converge broadly into two groups, one in which firms hold few assets and find it hard to invest and grow out of their credit constraints because they are severely credit constrained, and a group of older firms that hold a large amount of assets and are relatively unconstrained. A neg-

⁸At the heart of the general equilibrium effects is a pecuniary externality in the investment duration choice. When firms decide how much to invest in the long-term project, they do not internalize the impact of their choices on future interest rates and thereby on current and future tightness of borrowing constraints and investment choices of other firms. In this sense, this externality generates a dynamic macroeconomic complementarity: the anticipation of low output and high interest rates in the future feeds back into low long-term investment in the present and low future output. A similar mechanism in the context of entrepreneurial risk-taking is identified by Angeletos and Calvet (2006).

ative financial shock hurts credit constrained firms' total investment, which was already relatively small in size before the shock, and produces a moderate drop in output, but a positive shock enables these firms to overcome their constraints after a few periods and significantly increase their investment, resulting in a large output increase. We are able to capture this strong asymmetry in the response to financial shocks because our framework allows for credit constraints to bind occasionally, considers firm heterogeneity along two important dimensions, and uses a solution method that can capture non-linearities in the response to shocks. These features, in addition to investment composition choice, thus seem important when constructing models designed to study the macroeconomic implications of credit constraints.

A large body of research has studied the role of firms' financing frictions in amplifying business cycles. Most of this work focuses on how firms' investment capacity is affected by tighter borrowing constraints in recessions or following a tightening of monetary policy, either directly through a balance sheet channel (Bernanke and Gertler (1989), Kiyotaki and Moore (1997), Carlstrom and Fuerst (1997), Bernanke at al. (1999)) or indirectly through a contraction in the supply of intermediated finance (Holmstrom and Tirole (1997), Gertler and Kiyotaki (2010)). Common to most of these models is the assumption that firms can invest in only one type of project, which precludes analyzing questions regarding the composition of investment. In addition, these models typically assume permanently binding credit constraints, which, in combination with the assumption of lack of investment type choice, means that the effect of the anticipation of future constraints is limited to general equilibrium effects that affect entrepreneurs through asset or product prices. Entrepreneurs who have an incentive to insure against future credit constraints are unable to do so because their possible actions are limited to investing as much as their permanently binding constraints allow in their single investment opportunity.

Matsuyama (2007) and Aghion et al. (2010) are two exceptions in this literature. They both address the behavior of the composition of investment in overlapping generations models with homogeneous entrepreneurs that suffer from financial constraints, although they mostly focus on questions related to economic growth. Matsuyama (2007) considers one-period investment projects that differ exogenously in their investment size, the pledgeability of their returns, and their productivity, and uses the framework to explain a wide range of phenomena related to economic growth and development. The focus in our paper instead is on short-run aggregate dynamics and on duration as a primary source of investment project heterogeneity. Furthermore, in our paper differences in duration deliver endogenous and time-varying differences in pledgeability, which is assumed exogenously in their work, and liquidity risk, which is absent. Aghion et al. (2010) on the other hand assume that entrepreneurs can invest in a short-term project that takes one period to complete and in a long-term project that takes two periods to complete, suffers from liquidity risk and has a positive externality on the productivity of future generations. Financial frictions discourage investment in long-term projects and tend to make them procyclical, and this is the source of a mechanism that can successfully explain the documented negative correlation between mean growth and volatility. Common to our paper is the notion that long-term projects generate more liquidity risk. But our paper differs in several key modeling choices, in that it tries to develop a more realistic framework amenable to quantitative evaluation and comparison to the data, and in its main results. To highlight some of the most relevant differences, first, the short run implications in their model, that more severe credit frictions amplify the response of output to exogenous business cycle shocks, is the opposite to ours, in which more severe credit frictions dampen the response.⁹Second, one key modeling choice distinction is that, in their model, long-term projects (which could be also interpreted as riskier, rather than longer-term, projects) are assumed to be more productive and to deliver this high productivity fast. Third, this modeling choice has another important implication, which is that in their model firms do not accumulate long-term capital. In our model instead many firms hold large stocks of long-term capital, which by its nature is totally or partially irreversible and depreciates slowly, and this has a large impact on the short-run reaction of long-term investment to shocks. Fourth, in their model the amplification and propagation effects operate entirely through a pure externality. While this is surely an important channel, one would imagine that the most important effects work through future output and productivity changes that the firm internalizes. Finally, relative to both Matsuyama (2007) and Aghion et al. (2010), our paper introduces long-lived firms and allows for firm heterogeneity in the dimensions of asset holdings and long-term capital holdings, enabling us to do a realistic quantitative evaluation of the importance of our proposed mechanism.¹⁰

Long-term investment in our paper is modeled as the production of units of a capital good that is firm-specific. Firm-specificity of capital introduces irreversibility in investment, and relates our results to the literature studying how credit constraints interact with investment irreversibility to exacerbate aggregate fluctuations. Caggese (2007) studies a partial equilibrium setup in which firms invest in variable and fixed capital, and there is total irreversibility in fixed investment. Khan and Thomas (2011) introduce a dynamic stochastic general equilibrium framework to study the interaction of collateral constraints and partial irreversibility of investment, but do not address issues of composition of investment.

⁹Comparing this result is complicated by the fact that they do not perform a quantitative analysis of their model, but the differences are not a matter of timing; in our model, dampening lasts for around 15 quarters, relative to an economy with no credit frictions, which means that the result is not limited to very high frequency dynamics.

¹⁰Introducing firm-level heterogeneity allows us, amongst other things, to generate a realistic distribution of the severity of firms' credit constraints and to capture the possibility that, in a general equilibrium context, financially unconstrained firms react in such a way that they compensate for constrained firms' behavior and aggregate dynamics are hardly affected by the presence of financing constraints.

Finally, the corporate finance literature has studied how financial constraints may impact aspects of firms' investment other than the quantity of investment. While most of this work is empirical¹¹, some theoretical papers have also addressed this topic. In Almeida et al. (2011), firms that expect future financing constraints prefer to invest in more liquid, shorter-term and safer assets. In Eisfeldt and Rampini (2007), firms that suffer from credit constraints have a preference for used capital, which requires a lower up-front investment than new capital.

The remainder of the paper is organized as follows. Section 2 introduces the model. Section 3 discusses the calibration of the model and analyzes firms' optimal decisions in the steady state. Section 4 presents the main results of the model concerning the reaction of the economy to productivity shocks, and section 5 extends the analysis to financial shocks and asymmetries in the response to shocks. Section 6 concludes.

2 Model

We introduce an infinite-horizon, discrete time, perfect foresight model of an economy populated by a continuum of identical households and a continuum of financially constrained firms, both of measure one. There are three goods in this economy; short-term capital, long-term capital, and a homogenous final good that can be used for consumption and investment. Firms are heterogeneous along two dimensions; their holdings of the final good, and their holdings of long-term capital. We begin our description of the economy with a detailed study of the optimization problem facing each firm, then follow with a brief discussion of households and equilibrium.

2.1 Firms

Production opportunities

We assume a large number of firms, each producing the final good via two production processes, both of which use capital (short-term or long-term) and labor.

A firm can invest in a *short-term* production opportunity that converts $k_{S,t+1}$ units of short-term capital into units of the final good next period using the technology

$$f_S(k_{S,t+1}) = \theta_{S,t+1} k_{S,t+1}^{\alpha_S},$$

¹¹Examples of empirical papers are Chevalier and Scharfstein (1996), Andrade and Kaplan (1998), Peyer and Shivdasani (2001), Ahn et al. (2006), Eisfeldt and Rampini (2007) and Aghion et al. (2012).

where $\theta_{S,t+1}$ is an aggregate productivity term given by

$$\theta_{S,t+1} = \theta_S \theta_{t+1}$$

and where θ_S is a constant productivity parameter of the short-term technology and θ_{t+1} is an aggregate productivity factor common to both short and long-term production technologies. Investment in the short-term technology depreciates fully in one period.

A firm can also invest in a *long-term* production opportunity that converts $k_{L,t+1}$ units of long-term capital into units of the final good next period using the technology

$$f_L(k_{L,t+1}) = \theta_{L,t+1} k_{L,t+1}^{\alpha_L}$$

where $\theta_{L,t+1}$ is an aggregate productivity term given by

$$\theta_{L,t+1} = \theta_L \theta_{t+1}$$

and where θ_L is a constant productivity parameter of the long-term technology. Long-term capital is firm-specific and depreciates at a rate of δ_L , where $0 < \delta_L < 1$. Accumulation of long-term capital follows the rule

$$k_{L,t+1} = (1 - \delta_L)k_{L,t} + i_{L,t},\tag{1}$$

where $i_{L,t}$ represents period t investment in long-term capital. A firm can choose not to utilize all long-term capital in a given period, in which case the unused part immediately depreciates. The possibility of disinvestment is captured by the condition that

$$i_{L,t} \ge -(1 - \delta_L)k_{L,t}.\tag{2}$$

Positive investment in long-term capital suffers from convex costs of adjustment according to the function $c(i_{L,t})$, which satisfies c(0) = 0, $c'(\cdot) > 0$ and $c''(\cdot) > 0$.¹² Not utilizing part of the capital stock in a given period is equivalent to disinvesting, with an endogenous cost equal to the value for the firm of those units of capital not being utilized.¹³

Firms face the possibility of suffering a liquidity shock s_t each period with probability p. The size of the liquidity shock is a function of the stock at the end of a period of each

¹²Even though adjustment costs are not modeled for short-term investment, this is not because of the belief that these costs are not relevant for short-term investment. It is because in our model short-term investment depreciates fully within one period and one can introduce them by adjusting the short-term production function accordingly. In the interest of simplicity we choose then not to model them separately.

¹³Adjustment costs of investment are introduced because the macroeconomic dynamic stochastic general equilibrium (DSGE) literature has found that they are an important feature in bringing the performance of these models closer to the data (Smets and Wouters (2007)).

type of capital, according to the function

$$s_t = \varepsilon \left(k_{S,t+1} + k_{L,t+1} \right), \tag{3}$$

where $\varepsilon > 0$. The occurrence of the liquidity shock is captured by the indicator function 1_s , which takes value 1 if the liquidity shock occurs, and value 0 if it doesn't.

Firms need one unit of labor to operate irrespective of the scale, in exchange for a wage w_t that is paid at the beginning of the period. Given that revenues are not received until the following period, the wage bill needs to be financed with internal or external funds. The wage is determined according to the sharing rule

$$w_t = (1 - \alpha_S)\theta_{S,t+1}k_{S,t+1}^{\alpha_S} + (1 - \alpha_L)\theta_{L,t+1}k_{L,t+1}^{\alpha_L}.$$
(4)

Firms' financing

Firms need to finance wages and investment, including adjustment costs. They can do so by using retained earnings or by borrowing. Firms are able to borrow an amount b_t (or save an amount $-b_t$) using one-period debt contracts at the riskless interest rate $1 + r_{t+1}$.

A firm's holdings of the final good, which we will also refer to as "asset holdings" or "asset position", computed at the beginning of a period immediately after dividends for that period are paid, is denoted $a_{f,t}$, and the dynamics of this asset position are given by

$$a_{f,t+1} = f_S\left(k_{S,t+1}\right) + f_L\left(k_{L,t+1}\right) - d_{t+1} - b_{t+1}(1+r_{t+1}) - 1_s s_t,\tag{5}$$

where d_{t+1} are dividends paid to shareholders, $k_{L,t+1}$ is given by expression (1), and b_{t+1} can be expressed, using the budget constraint for a firm, as

$$b_{t+1} = i_{L,t} + c(i_{L,t}) + k_{S,t+1} + w_{t+1} - a_{f,t} \quad \text{if } i_{L,t} \ge 0, \tag{6}$$

or

$$b_{t+1} = k_{S,t+1} + w_{t+1} - a_{f,t} \quad \text{if } i_{L,t} < 0.$$
(7)

We will denote $a_{f,t+1}^-$ the asset position that results in the case that the liquidity shock occurred, and conversely $a_{f,t+1}^+$ the asset position in the case it did not.

Firms face financing constraints that limit their ability to obtain external finance using both debt and equity. Firms are unable to obtain any equity finance, which means that their dividends cannot be negative, so

$$d_t \ge 0. \tag{8}$$

Not being able to issue equity also means that firm assets can never be below zero, or

$$a_{f,t} \ge 0,\tag{9}$$

because, should the firm be forced to exit (which occurs with positive exogenous probability, as will be discussed next) with a negative asset position, the shareholders would have to contribute additional funds to cover the negative asset position. This would violate constraint (8).

We assume that firm shareholders force firms to pay out dividends every period according to the rule

$$d_{t+1} = \max\left\{\tau\left(f_S\left(k_{S,t+1}\right) + f_L\left(k_{L,t+1}\right) - b_{t+1}(1+r_{t+1}) - 1_s s_t - a_{f,t}\right), 0\right\},\tag{10}$$

which establishes that a fixed fraction of earnings, if these are positive, have to be paid out to shareholders.

Firms also face a borrowing limit given by:

$$b_{t+1} \le \lambda_t \frac{f_S(k_{S,t+1}) + f_L(k_{L,t+1}) - s_t}{1 + r_{t+1}} \tag{11}$$

which means that they can borrow up to a fraction λ_t of the present discounted value of the lowest possible earnings next period, which occur if the firm suffers a liquidity shock.¹⁴ Earnings obtained in periods after t+1 cannot be pledged by committing to roll over the debt because, as will be explained next, firms cannot commit to be in operation the following period.¹⁵

Firm exit

Firms face an exogenous exit shock which occurs with probability η , which can be interpreted as a permanent negative idiosyncratic productivity shock that forces firms to exit. Firms distribute all of their asset holdings $a_{f,t}$ as dividends before exiting but their

¹⁴It is also important to note that b_{t+1} is debt net of asset holdings given how it is defined. Asset holdings and their returns could in principle be pledgeable, and if that is the case then firms could hold a positive balance of assets in their balance sheet and simultaneously be borrowing. If debt is not risky, however, as is the case in this model, firms have no motive to do so. See Acharya et al. (2007) for a detailed discussion of this point.

¹⁵This specification for borrowing constraints can have several microeconomic justifications. As a first step, if we restrict the possible financial contracts to riskless debt, then it follows that only next period's output can be pledgeable. This is because firms might be forced to exit next period and thus cannot roll over debt and pledge output occurring in t + 2 or after. One needs then to justify why only a fraction of this output is pledgeable. Following Hart and Moore (1994) and Kiyotaki and Moore (1997), one could assume that each project requires the input of the firm manager and that without her services revenues are a fraction $(1 - \lambda_t)$ lower. The she can threaten to withdraw her services and renegotiate her liability down to a fraction λ_t of revenues. Another justification could be along the lines of Holmstrom and Tirole (1997). One could assume that the borrower has to contribute with an unobservable effort to increase the expected return of the project and that in order for her to have the proper incentives at least a particular fraction of the expected returns need to accrue to her and cannot be pledged to lenders.

long-term capital holdings $k_{L,t}$ lose all of their value as they are firm-specific. A firm that exits is immediately replaced by a new firm with no assets or long-term capital.

Firm Optimization

Firms maximize the present discounted value of dividends d_t distributed to their shareholders, the households, with perfect foresight on the path of future interest rates, aggregate productivity and financing constraints. Given a sequence starting at time t of the interest rate, aggregate productivity and financing constraints $z_t = \{r_s, \theta_s, \lambda_s\}_{s=t}^{s=\infty}$, a firm's value function, calculated at the beginning of period t conditional on continuing to operate (so assuming it has not suffered the exogenous exit shock), and immediately after the dividend payment, is given by

$$J(a_{f,t}, k_{L,t}, z_t) = \max_{\substack{d_{t+1}, k_{S,t+1}, k_{L,t+1}, b_{t+1}, w_t}} M_{t,t+1} \left[d_{t+1} + \eta \left(p a_{f,t+1}^- + (1-p) a_{f,t+1}^+ \right) + (1-\eta) \left(p J(a_{f,t+1}^-, k_{L,t+1}, z_{t+1}) + (1-p) J(a_{f,t+1}^+, k_{L,t+1}, z_{t+1}) \right) \right],$$

subject to (5), (4), (8), (10), and (11), where $M_{t,t+1}$ is the discount factor of households that the firm uses to discount future dividends.

Firms' optimal choices for short term investment, long term capital holdings, wages, dividends and borrowing are given by $k_{S,t+1}(a_{f,t}, k_{L,t}, z_t), k_{L,t+1}(a_{f,t}, k_{L,t}, z_t), b_{t+1}(a_{f,t}, k_{L,t}, z_t), w_t(a_{f,t}, k_{L,t}, z_t)$, and $d_{t+1}(a_{f,t}, k_{L,t}, z_t)$.¹⁶

2.2 Households

There is a continuum of measure one of infinitely lived households. Given that there is perfect insurance between households, we will treat the household sector as a representative agent. Let $\Gamma_t(a_f, k_L)$ denote the joint distribution of asset holdings and long-term capital holdings in the population of firms, of which there is a continuum also of measure one. Households have preferences represented by the utility function

$$\sum_{t=0}^{\infty} \beta^t u(C_t), \tag{12}$$

¹⁶It is important to point out that in the context of this model and in the absence of any assumptions that could influence dividend payout behavior (such as the one we make in (10)), an optimizing firm would delay dividend payments until it is forced to exit. This is because firms use households' intertemporal marginal rate of substitution, can save at the same interest rate as households, and face borrowing constraints, which means that holding liquidity inside the firm to avoid potential future underinvestment is more valuable than distributing it out to shareholders. Firms in practice pay out dividends regularly and in large amounts, and this is the justification for our assumption (expression (10)) from an empirical viewpoint. From the perspective of theory, one could justify this assumption on the basis of Jensen's (1986) hypothesis that dividends reduce agency costs associated with free cash flow.

where C_t is consumption in period t. They are endowed with one unit of labor services which they provide inelastically to the firms in exchange for wage W_t equal to

$$W_t = \int w_t(a_{f,t}, k_{L,t}, z_t) d\Gamma_t(a_f, k_L).$$
(13)

Households can borrow or lend at the riskless interest rate r_{t+1} and they own shares in the firms, which give them the right to any dividends distributed by the firms, which are equal to

$$D_t = \int d_t(a_{f,t}, k_{L,t}, z_t) d\Gamma_t(a_f, k_L).$$

Their budget constraint for period t is given by

$$C_t = N_t + W_t - A_{t+1}, (14)$$

where

$$N_t = (1+r_t)A_t + D_t$$

is the wealth of a consumer at the beginning of period t, and A_{t+1} are savings made in period t.

2.3 Equilibrium

Given a sequence starting at time t of aggregate productivity and borrowing constraints which we denote by $x_t = \{\theta_s, \lambda_s\}_{s=t}^{s=\infty}$, let firms' choices for short term investment, long term capital holdings, and borrowing be given by $k_{S,t+1}(a_{f,t}, k_{L,t}, x_t), k_{L,t+1}(a_{f,t}, k_{L,t}, x_t)$, and $b_{t+1}(a_{f,t}, k_{L,t}, x_t)$, respectively, and choices by households for consumption and savings be given by $C_t(N_t, x_t)$ and $A_{t+1}(N_t, x_t)$, respectively. Let $\Gamma_t(a_f, k_L)$ denote the joint distribution of asset holdings and long-term capital holdings in the population of firms. We are now ready to define an equilibrium.

Definition 1 An equilibrium is a sequence of interest rates $\{r_t\}$, a sequence of consumption and savings policies $\{C_t(N_t, x_t)\}$ and $\{A_{t+1}(N_t, x_t)\}$, a sequence of short term investment, long term capital holdings, and borrowing policies $\{k_{S,t+1}(a_{f,t}, k_{L,t}, x_t)\}$, $\{k_{L,t+1}(a_{f,t}, k_{L,t}, x_t)\}$, and $\{b_{t+1}(a_{f,t}, k_{L,t}, x_t)\}$, and a sequence of distributions for firms' asset and long-term capital holdings $\{\Gamma_t\}$, such that, given the initial distribution Γ_0 and an initial wealth of households N_0 :

(i) $C_t(N_t, x_t)$ and $A_{t+1}(N_t, x_t)$ are optimal given $\{r_t\}$,

(ii) $k_{S,t+1}(a_{f,t}, k_{L,t}, x_t)$, $k_{L,t+1}(a_{f,t}, k_{L,t}, x_t)$, and $b_{t+1}(a_{f,t}, k_{L,t}, x_t)$ are optimal given $\{r_t\}$,

(iii) Γ_t is consistent with the investment and borrowing decisions of firms,

(iv) the bond market clears:

$$\int b_{t+1}(a_{f,t}, k_{L,t}, x_t) d\Gamma_t(a_f, k_L) = A_{t+1}(N_t, x_t).$$
(15)

The bond market clearing condition (15) requires that net aggregate borrowing of the firm sector (the left-hand side of the equation) is equal to aggregate saving of the household sector (the right-hand side).

3 Calibration and Steady State Analysis

3.1 Calibration

We calibrate the economy at the quarterly frequency, and the chosen values for the parameters are shown in Table 1.

The utility function for households is chosen to be iso-elastic of the form

$$u(C_t) = \frac{C_t^{1-\gamma} - 1}{1-\gamma},$$

with a constant relative risk aversion coefficient, γ , of 2. The intertemporal preference rate β is set at 0.99 to match a 4 percent annualized average interest rate.

The calibration of parameters concerning the firm sector can be divided into three categories: those that affect their investment opportunities, those that control firms' financing constraints, and those that determine firm entry and exit.

The firm technology parameters are calibrated as follows. The capital share of both technologies, α_S and α_L , is set to 0.36, consistent with most macroeconomic studies. There are four remaining parameters affecting firms' technologies. These are the productivities of the short-term and long-term technologies, θ_S and θ_L respectively, the probability p of suffering a liquidity shock, and the size ε of the liquidity shock per unit of capital. These four parameters are jointly calibrated to match four moments in the data: (i) the aggregate amount of long-term investment as a share of total investment, (ii) the ratio between the median asset holdings of young firms and the median asset holdings of the whole population of firms, (iii) the standard deviation of the ratio of profits to sales, and (iv) the probability of negative profits. The ratio of long-term to total investment is calculated for the U.S. using data from the Bureau of Economic Analysis (BEA). The BEA provides data on gross private domestic investment, which includes fixed residential and nonresidential investment and changes in private inventories, and also on research and development (R&D) spending. Long-term investment is considered to include non-residential fixed investment in structures, residential fixed investment, and R&D spending, and short-term investment includes fixed investment in equipment and

Table	1:	Parameter	Values
-------	----	-----------	--------

Parameters	Explanation	Value
γ	Coefficient of relative risk aversion	2
β	Households' discount rate	0.99
α_S	Capital share in short-term technology	0.36
α_L	Capital share in long-term technology	0.36
heta	Aggregate productivity factor (steady state)	1
θ_S	Productivity factor of short-term technology	0.7
θ_L	Productivity factor of long-term technology	0.8
p	Probability of a negative liquidity shock	0.25
ε	Size of liquidity shock per unit of total capital	3
δ_L	Depreciation of long-term capital	0.025
ζ	Multiplicative parameter of convex adjustment cost function	400,000
χ	Exponential parameter of convex adjustment cost function	4
λ_{SS}	Pledgeability of next period's earnings	0.8
au	Dividend payout ratio	0.975
η	Exogenous probability of firm exit	0.025

software and changes in private inventories. The share of long-term investment using this measure has been on average 55% during 1990 - 2007. The ratio between the median asset holdings of young stock-exchange listed firms (those with less than 1 year following their IPO) and the median asset holdings of the whole population of listed firms is 0.47 in the U.S., using data from Capital IQ-Compustat for 2002-2011. Using this same data, we get a standard deviation of the ratio of profits to sales of 0.35 and a probability of negative profits of 25%, both calculated at the annual frequency. A summary of all the empirical moments used to calibrate parameters can be found in Table 2.

The depreciation rate of long-term capital is set at $\delta_L = 0.025$, which is a standard value for quarterly Real Business Cycle (RBC) models. The adjustment cost function for long-term investment, $c(i_{L,t})$, is assumed to adopt the functional form

$$c(i_{L,t}) = \chi i_{L,t}^{\zeta},$$

with values $\zeta = 4$ and $\chi = 400,000$ set to generate aggregate adjustment costs that are roughly equal to 0.2% of aggregate investment, in the lower end of the estimates discussed in studies of adjustment costs of capital such as Gourio and Kashyap (2007), and also to be broadly in line with the firm-level standard deviation of investment rates (investment over capital) observed in the data. Cooper and Haltiwanger (2006) show using Longitudinal Research Database data between 1972 and 1988 that the standard deviation of the investment rate is 0.306.

The steady state value for the parameter regulating borrowing constraints (λ_{SS}) is set to match the average net leverage ratio for Compustat publicly listed firms, which is

Table 2:	Empirical	Moments	Matched	in	the	Calibration
----------	-----------	---------	---------	---------------	-----	-------------

Empirical Moment	Source	Data	Model	
Targets to calibrate θ_L, θ_S, p and ε				
- Long-term investment over total investment	(1)	0.55	0.62	
- Median assets of young firms over median assets of all firms	(2)	0.47	0.52	
- Probability of negative profits (annual)	(2)	0.25	0.33	
- Standard deviation of the ratio of profits to sales (annual)	(2)	0.35	0.27	
-	()			
Targets to calibrate ζ and χ				
- Adjustment costs as share of aggregate investment	(3)	0.2% - 6.2%	0.2%	
- Standard deviation of investment over capital (annual)	(4)	0.306	0.175	
Targets to calibrate λ				
- Average net leverage ratio	(5)	0.079	0.068	
	. ,			
Targets to calibrate $ au$				
- Average payout ratio of U.S. corporations	(6)	105% - 116%	85%	
Targets to calibrate η				
- Job destruction rate (annual)	(7)	10%	10%	

(1) Own calculations using Bureau of Economic Analysis (BEA) data, (2) Own calculations using Capital IQ-Compustat for U.S. listed firms, (3) Gourio and Kashyap (2007), (4) Cooper and Haltiwanger (2006), (5) Bates et al. (2009), (6) Weston and Siu (2003), (7) Bilbiie et al. (2012)

7.92% according to Bates et al. (2009) using data from 1980 to 2006. They calculate net leverage as the ratio of total debt minus cash holdings to the book value of total assets. Net leverage in our model is calculated as the current value of debt b_{t+1} minus holdings of the asset over the total value of book assets, assuming assets are marked to market and valued according to the current firm value $J(a_{f,t}, k_{L,t}, z_{SS})$, where z_{SS} is the steady state defined by $\{r_{SS}, \theta_{SS}, \lambda_{SS}\}$.¹⁷ The payout ratio for firms with positive earnings (τ) is set to be roughly in line but somewhat below the average payout ratio of U.S. corporations including share repurchases, which Weston and Siu (2003) show was between 105% and 116% in the period 1998-2001. A lower payout ratio compensates for the fact that our firms are assumed not to be able to issue new equity.

Finally, we set the size of the exogenous firm exit shock $\eta = 0.025$ to match the U.S. empirical level of 10 percent job destruction per year, following Bilbiie et al. (2012). Exiting firms are immediately replaced by new firms with no holdings of assets or long-term capital.

¹⁷Net leverage and leverage are the same in our model because firms do not simultaneously hold the asset and borrow. For this reason, the relevant counterpart in the data is net leverage.



Figure 1: FIRM VALUE AND THE COMPOSITION OF INVESTMENT. The top figure displays the net value of the firm in the space of asset holdings $a_{f,t}$ and long-term capital holdings $k_{L,t}$. The net value of the firm is calculated as firm value $J(a_{f,t}, k_{L,t}, z_t)$ minus asset holdings. The bottom figure displays long-term investment as a share of total investment in the same space defined by asset holdings and long-term capital holdings.

3.2 Firm Policies and Firm Heterogeneity in the Steady State

We now take a closer look at firm policies in the steady state equilibrium that arises from the calibration described above. We solve the model numerically using value function iteration and simulation techniques. A detailed explanation of the solution method for the steady state equilibrium can be found in the Appendix.

The top graph of Figure 1 shows the net value of the firm in the space of asset holdings $a_{f,t}$ and long-term capital holdings $k_{L,t}$. The net value of the firm is calculated as firm value $J(a_{f,t}, k_{L,t}, z_t)$ minus asset holdings. Net value is increasing and concave in both arguments, and drops sharply as asset holdings approach zero. The strong concavity of the value function with respect to asset holdings introduces risk aversion and precautionary behavior in firms that affects their investment portfolio choices. These choices



Figure 2: FIRM BORROWING CONSTRAINTS AND SPARE DEBT CAPACITY. The top figure displays the endogenous borrowing constraint of firms in the space of asset holdings $a_{f,t}$ and long-term capital holdings $k_{L,t}$. The lower figure shows the spare debt capacity, which is calculated as the difference between the borrowing constraint and actual borrowing, in the same space defined by asset holdings and long-term capital holdings.

are captured in the bottom graph of Figure 1, which displays long-term investment as a share of total investment in the same space defined by asset and long-term capital holdings. Long-term investment decreases as a share of total investment as asset holdings decrease. This is the result of two forces. On one hand, long-term investment generates less pledgeable output as a share of its total future output than short-term investment. Only next period returns are potentially pledgeable, and long-term investment generates most of its returns further in the future. This means that credit constrained firms can leverage long-term investment less than short-term investment, and this feature makes short-term investment relatively more interesting for a credit constrained firm. On top of this, long-term investment generates liquidity risk, which credit-constrained firms are keen to avoid. The increase in liquidity risk comes about because the size of the liquidity



Figure 3: FIRM HETEROGENEITY. This figure displays the distribution of asset holdings $a_{f,t}$ and long-term capital holdings $k_{L,t}$ in the steady state equilibrium. The benchmark economy in which the pledgeable fraction of next period's revenues in the steady state is $\lambda_{SS} = 0.8$ is compared with an economy with more severe credit constraints in which $\lambda_{SS} = 0.7$.

shock is proportional to total capital (long and short-term), and long-term capital depreciates slowly. Increases in long-term investment today increase liquidity risk for several periods into the future. For these two reasons, firms with low asset holdings who are facing credit constraints or are likely to face them in the near future shift from long-term to short-term investment. Long-term investment also decreases with long-term capital holdings, but only very gradually given the convex costs of adjusting long-term capital.

The financing constraints and borrowing decisions of firms are captured in Figure 2. The top graph displays the endogenous equilibrium borrowing constraints of firms in the space of asset and long-term capital holdings. The maximum amount a firm can borrow is equal to a fraction $\lambda_{SS} = 0.8$ of its lowest possible output next period (the state in which it has suffered a liquidity shock). The relationship between long-term capital holdings and borrowing constraints is non-linear. Long-term capital generates output next period which is partially pledgeable, but on the other hand increases the size of the liquidity shock. The first effect dominates for low levels of long-term capital when the average return to capital becomes low. The relationship between assets and equilibrium borrowing constraints is more straightforward. Low asset holdings enable low investment and low pledgeable output. In the bottom graph we observe the actual borrowing behavior of firms. It displays the spare debt capacity of firms; firms with no spare debt capacity suffer binding credit constraints. Firms with low asset holdings borrow up to capacity, while firms with higher levels of asset holdings have spare capacity.

Figure 3 displays the distribution of asset and long-term capital holdings, the two dimensions along which firms differ. To better understand the impact of financing constraints on firm heterogeneity, we include the equilibrium distribution of firms for the benchmark calibration and for one that features tighter borrowing constraints, captured by a fraction of next period's output that can be pledged, λ_{SS} , of 70%, instead of 80% as in the benchmark calibration. It is clear from the figure that borrowing constraints slow down both asset and long-term capital accumulation. The pattern of asset holdings shows that firms cluster around two groups forming a twin-peaked distribution. Firms with very low asset holdings, which are either young firms, or older firms that have suffered several negative liquidity shocks, are unable to invest large quantities and their assets grow very slowly. They face very tight borrowing constraints as was shown in Figure 2. Beyond a level of asset holdings, they are able to leverage their existing assets more and thus invest more and accumulate assets more quickly. Given the decreasing marginal return to both types of capital, eventually returns to investment decrease and so does asset accumulation, and another large grouping of firms arises with large asset holdings.

4 Amplification and Propagation of Productivity Shocks

In this section we analyze the transitional dynamics following an unexpected, temporary and persistent negative shock to aggregate productivity θ_t . We solve the model numerically using value function iteration and simulation techniques. A detailed explanation of the solution method for the transitional dynamics can be found in the Appendix.

We introduce a shock to aggregate productivity θ_t equivalent to -1% of its steady state value ($\theta = 1$) in t = 1. We assume that following the shock, θ_t evolves according to

$$\log \theta_t = \rho \log \theta_{t-1}$$

where $\rho = 0.9$. Agents are assumed not to learn in advance about the shock.¹⁸ They have however a perfect foresight about the evolution of θ_t from then on. The exposition that follows is designed so that the different mechanisms in operation are clearly distinguished. We will first analyze an economy as described in the model of section 2, but in which the interest rate is assumed fixed at its steady state level throughout the transitional dynamics. This version with constant interest rates can be interpreted as a small open economy with perfect capital mobility in which the international interest rate is the one that holds in the steady state. This version of the model will allow us to study the direct

¹⁸In other words, they do not know, in period t when households and firms are making their optimal choices about consumption, savings, investment and borrowing, that a shock will hit the economy in period t+1 that will affect the return on savings and the output that results from this period's investment.



Figure 4: RESPONSE TO A NEGATIVE AGGREGATE PRODUCTIVITY SHOCK: FIXED INTER-EST RATE CASE. The figure displays the response of nine key macroeconomic variables to a negative and persistent shock to aggregate productivity (TFP) θ_t equal to -1% of its steady state value $\theta_{SS} = 1$. The interest rate r_t is assumed to be fixed at its steady state value $r_{SS} = 1.01\%$. The responses of a model featuring frictionless credit markets are compared with those of one featuring credit constraints. The responses of all variables except the TFP shock, the interest rate, and long-term investment (share) are calculated as percent deviations from their steady state values. The response of the share of long-term investment is calculated as the difference between the share in each period and its steady state value.

impact of productivity changes on firms' investment choices. Next, we will return to the exact description of the economy described in section 2 in which the interest rate is endogenous to analyze the general equilibrium interactions that arise in the transitional dynamics. In both cases, we will compare the response of the benchmark economy with financing frictions with one in which financing frictions are absent.

4.1 Fixed Interest Rate

The responses of nine key aggregate variables to the productivity shock in the version of the model with no general equilibrium effects are shown in Figure 4. In that figure, the response of the benchmark economy with financial constraints and the response of an economy with financially unconstrained firms is compared.¹⁹

In the economy with no financial constraints, we find that both short and long-term investment drop moderately following the shock. Short term investment then recovers slowly towards the steady state level, while long term investment recovers faster and rises for several periods above the steady state level. Two forces are at play. The decrease in productivity discourages investment in both short and long-term investment, all else equal. On impact the effect is stronger for long-term investment however because most firms already hold some long-term capital, which depreciates slowly. The drop in shortterm investment on the other hand is gradual and hump-shaped. The other force at play has to do with the change in the relative returns of long-term versus short-term investment. A temporary negative productivity shock decreases short term investment returns more than long-term investment returns, and this acts to encourage long-term investment as a share of total investment. This second force explains the more rapid recovery of long-term investment. Put together, this means that long-term investment as a share of total investment falls below the steady state level initially to rise above that level subsequently and stay high for around 20 quarters. The level of aggregate long-term investment eventually also rises above its steady state level after around 10 quarters. The effect on output and households' consumption is moderate, with both quantities roughly tracking the productivity process with little amplification or propagation.

The behavior of the composition of investment when firms face financing constraints changes substantially. For the initial quarters following the shock, long-term investment drops significantly. It falls by 7.66% relative to the steady state level in the first quarter, compared to only 4.42% in the unconstrained case. Short-term investment instead falls much less than in the unconstrained case. It bottoms at -0.35% in the third quarter, while it falls three times as much as (-1.11%) in the unconstrained case. Firms, facing a worsening of their financing constraints, react by cutting strongly on long-term investment while keeping short-term investment. As a result, the share of long-term investment in total investment falls by 6.4 percentage points relative to the steady state share, compared to 3 percentage points in the unconstrained case. This translates into a large drop in the stock of long-term capital which takes many periods to recover and propagates the effects of the shock.

Firms' investment dynamics translate into a dampening of the effects of a negative productivity shock in the short-run, and a moderate propagation in the medium and longrun. Output drops more and sooner in the unconstrained case; it bottoms at -1.02%relative to the steady state level in the fourth quarter following the shock, while the output drop in that quarter when firms face financing constraints is -0.79%. In the

¹⁹The calibration is kept unchanged for both versions of the model. The responses plotted in Figure 4 as deviations from steady state values are calculated relative to the relevant steady state for each case.



Figure 5: RESPONSE TO A NEGATIVE AGGREGATE PRODUCTIVITY SHOCK: ENDOGENOUS INTEREST RATE CASE. The figure displays the response of nine key macroeconomic variables to a negative and persistent shock to aggregate productivity (TFP) θ_t equal to -1% of its steady state value $\theta_{SS} = 1$. The interest rate r_t is assumed to be endogenous. The responses of a model featuring frictionless credit markets are compared with those of one featuring credit constraints. The responses of all variables except the TFP shock, the interest rate, and long-term investment (share) are calculated as percent deviations from their steady state values. The response of the share of long-term investment is calculated as the difference between the share in each period and its steady state value.

medium and long run, the large initial drop in long-term capital and its slow recovery means that the propagation of the effects of the shock is greater in the economy with financing constraints.

4.2 Endogenous Interest Rate

The general equilibrium implications in our model are of particular relevance given that we introduce firm-level heterogeneity and calibrate the model to generate a realistic distribution of the severity of firms' credit constraints. This setup can capture the possibility that, in a general equilibrium context, financially unconstrained firms react in such a way that they compensate for constrained firms' behavior and aggregate dynamics are either hardly affected by the presence of financing constraints, or affected in a significantly different manner. The transitional dynamics in this case are displayed in Figure 5.



Figure 6: RESPONSE OF SELECTED FINANCIAL VARIABLES TO A NEGATIVE AGGREGATE PRODUCTIVITY SHOCK: ENDOGENOUS INTEREST RATE CASE. The figure displays the response of selected variables to a negative and persistent shock to aggregate productivity equal to -1% of its steady state value. The interest rate r_t is assumed to be endogenous. For some variables, the average values for firms in the lower and upper deciles of the distribution of asset holdings ("Small Firms" and "Large Firms", respectively) are displayed. The response of the share of long-term investment is calculated as the difference between the share in each period and its steady state value.

We focus again first on the economy in which credit market imperfections are absent to be able to compare it to the version in which they are present and understand their role. In this version, the interest rate r_t initially falls below its steady state level of 1.01% and subsequently rises above that level to peak at more or less 4% around 10 quarters following the shock. Interest rates then decrease very gradually back to their steady state level. This means that immediately following the shock, firms anticipate a sharp steepening of the yield curve. The decrease in interest rates in the short-run encourages short-term investment, while the steep yield curve discourages long-term investment. Eventually, interest rates rise and the slope of the yield curve decreases until it becomes downward sloping. The increase in short-run interest rates discourages both types of investment, but especially short-term investment, and the downward sloping yield curve encourages a shift towards long-term investment in the medium-term.

In the economy with financing constraints there are additional forces at play. Low interest rates in the early periods loosen borrowing constraints and encourage investment in both short and long term projects. Very high interest rates in the future however severely tighten borrowing constraints and induce a precautionary behavior in firms that encourages them to shift even more strongly into short-term investment. Both forces push short-term investment in the initial periods so that in equilibrium it is *above* the steady state level despite the decrease in productivity. Long-term investment dynamics are affected by yet another force. The interaction of convex costs of adjustment and borrowing constraints mean that firms have an incentive to spread out their investment in long-term capital. Anticipating very high interest rates and tight borrowing constraints that will restrict the amount that firms can invest in long-term they cut on long-term investment less following the shock than firms in the economy with no financing constraints.

The implications of firm' investment behavior for output and consumption are qualitatively similar as in the case with fixed r_t , but quantitatively significantly stronger. Output in the economy with no financial imperfections drops by as much as 2.6% relative to its steady state level in the seventh quarter, while it only drops as much as 1.8% in the eighth quarter in the economy with firm financing frictions. As it turns out, the general equilibrium forces enhance the financial decelerator effect and result in a further dampening in the short run of the effects of a temporary negative productivity shock.

4.3 Countercyclicality of financing constraints

It is worth inspecting more closely the dynamics of the severity of firm financing constraints and of firm heterogeneity. There are three mechanisms that influence how tight borrowing constraints are along the transitional dynamics. One is the decrease in firms' asset holdings following the shock. Another is the increase in interest rates which decreases the present value of pledgeable output and tightens borrowing constraints. These two sources of countercyclical financing constraints are standard and have been discussed at length in the literature.

A third, novel, source of countercyclicality of borrowing constraints arises from the coexistence of short and long-term investment opportunities. A temporary negative shock to productivity encourages a shift towards investing in long-term capital because its returns are relatively more acyclical. Long-term capital produces less pledgeable output per unit of investment, all else equal, so a shift towards long-term investment results in tighter borrowing constraints. Figure 6 shows some aspects of this mechanism. The share of borrowing constrained agents increases following a small initial drop. The initial drop can be explained by firms' lower demand for funds because of the combination of lower productivity and an existing large stock of long-term capital, and because of low interest rates. As long-term investment picks up and interest rates rise, so does the fraction of credit constrained firms. On the one hand, firms want to invest more in long-term capital in the medium-run (after around 10 quarters), which produces less pledgeable output



Figure 7: RESPONSE TO A NEGATIVE FINANCIAL SHOCK. The figure displays the response of six key macroeconomic variables to a negative shock to the severity of credit constraints (the pledgeability parameter λ_t) equal to -10%. The responses of models with a fixed and an endogenous interest rate r_t are displayed. Aggregate productivity (TFP) θ_t is assumed to be at its steady state value $\theta_{SS} = 1$. The responses of output and long-term capital are calculated as percent deviations from their steady state values. The response of the share of long-term investment is calculated as the difference between the share in each period and its steady state value.

than short-term investment and tightens borrowing constraints. Higher interest rates further decrease the present value of pledgeable output.

If we compare the behavior of small and large firms, which broadly capture financially constrained and unconstrained firms, respectively, we can observe that small firms react by strongly shifting their investment towards the short-term project while large firms do so to a much smaller extent. Small firms even *increase* their level of investment in the short-term project relative to the steady state level despite the decrease in productivity, because short-term investment produces more pledgeable output per unit of investment and thus requires a lower downpayment, and also as a precautionary reaction to the anticipation of future credit constraints.

5 Extensions

5.1 Amplification and Propagation of Financial Shocks

In this section we analyze the transitional dynamics following an unexpected, temporary and persistent negative shock to the severity of financing frictions. In particular, we introduce an unexpected shock to λ_t , the parameter that determines the share of next period's output that can be pledged to lenders today, equal to 10%. In other words, λ_t decreases in t = 1 unexpectedly from $\lambda_{t=0} = 0.8$ to $\lambda_{t=1} = 0.7$, and we assume that following the shock, for t > 1, λ_t evolves according to

$$\log \lambda_t = \rho \log \lambda_{t-1}$$

where $\rho = 0.6^{20}$ As with the technology shock, agents are assumed not to learn in advance about the shock. They have however a perfect foresight about the evolution of λ_t from then on. The aggregate productivity factor θ_t is assumed to remain constant at its steady state value θ_{SS} . The resulting transitional dynamics are displayed in Figure 7, in which both the fixed and endogenous r_t cases are shown.

In the case of a fixed r_t , a tightening of financing constraints results in lower total investment, which lowers output. It also translates into a lower share of long-term investment. Two forces are at play. Given that on the margin long-term investment produces less pledgeable output than short-term investment, a tightening of credit constraints shifts investment towards the short-term project. Given, on the other hand, that long-term investment increases future liquidity risk more than short-term investment, this interacts with firms' precautionary motive to also induce a shift towards the short-term project. Firms' precautionary behavior is also clear from the pattern of firms' borrowing. They decrease their borrowing to leave spare debt capacity, and this results in a decrease in the share of credit constrained firms. This increase in spare debt capacity comes about because the increase in the severity of financing frictions increases the concavity of the value function in the dimension of asset holdings and increases the precautionary motive of firms.

In the endogenous r_t case, the combination of lower output and the desire of households to smooth consumption translates into significantly higher interest rates for much of the transition period, following an initial brief decrease. As in the case with a technology shock, the term structure adopts a hump-shape with interest rates peaking at around 2.85% following five quarters. The effects on the dynamics of the composition of investment are small in the short run, but the effects on the level of investment and output are strong. Output drops by as much as 0.83% four quarters after the impact of the shock in the endogenous r_t case, compared to a maximum fall of 0.36% in the fixed r_t case in the third quarter. In the long-term, the large fall in the long-term stock of capital following the impact of the shock translates into a slow recovery, and much more so in the case with an endogenous interest rate. In that case, the stock of long-term capital and the level of output remain significantly below the steady state level for around 13

 $^{^{20}}$ The value for the quarterly persistence of the financial shock is taken from Hall (2011), who estimates the persistence of the increase in spreads during the financial crisis of 2008-2009.



Figure 8: ASYMMETRY IN THE RESPONSE TO SHOCKS. The figure displays the response of three key macroeconomic variables to negative and positive shocks to aggregate productivity (θ_t) and to the severity of financing constraints (λ_t). The response of the share of long-term investment is calculated as the difference between the share in each period and its steady state value.

quarters, even though 95% of the financial shock has died out by the 6th quarter.

5.2 Asymmetries

The nature of our solution method allows for the sign of the shock to influence transitional dynamics, and in Figure 8 we analyze the results of such an exercise by comparing the transitional dynamics of shocks to technology and financing frictions of different signs. We focus first on the analysis of technology shocks, in which we compare the reaction to a +1% and a -1% shock to aggregate productivity θ .²¹ The economy displays a moderately stronger reaction to negative technology shocks than to positive ones, but the differences are small. The difference is most obvious in interest rates, with a maximum fall of 232 basis points in the case of a positive shock, and a maximum increase of only 79 basis points in the case of a negative shock. The drop in long-term investment as a share of total investment in the negative shock case is nearly double the size as the increase in that ratio in the positive shock case. The differences in terms of the reaction of output are more moderate, with a maximum drop of 1.53% in the case of a negative shock case.

 $^{^{21}}$ For both shocks, we use the same persistence parameter values as in the analysis of section 4.

In the case of financial shocks, the differences are much more stark, and the shock that is most amplified is the positive shock (a loosening of financing frictions). In this exercise, we compare a +10% with a -10% shock to λ_t , the parameter that determines the share of next period's output that can be pledged to lenders today. In other words, λ_t decreases in t = 1 unexpectedly from $\lambda_{t=0} = 0.8$ to $\lambda_{t=1} = 0.7$ in one case, and increases in t = 1 unexpectedly from $\lambda_{t=0} = 0.8$ to $\lambda_{t=1} = 0.9$ in another. Output, the interest rate, and the share of long-term investment in total investment react much more strongly to a loosening of credit constraints than to a tightening. Output falls by 0.83% relative to its steady state value at the trough following a shock that increases financing frictions, but increases 2.88% at its peak following a credit constrained loosening.

The intuition for this large asymmetry is that in equilibrium firms converge broadly into two groups, one in which firms hold few assets and find it hard to invest and grow out of their credit constraints because they are severely credit constrained, and a group of older firms that hold a large amount of assets and are relatively unconstrained.²² A negative financial shock hurts credit constrained firms' total investment, which was already relatively small in size before the shock, and produces a moderate drop in output, but a positive shock enables these firms to overcome their constraints after a few periods and significantly increase their investment, resulting in a large output increase. This asymmetry in the response to a financial shock highlights the importance of allowing for credit constraints to bind occasionally and of carefully modelling firm heterogeneity in models that attempt to capture the macroeconomic implications of credit constraints.

 $^{^{22}{\}rm These}$ patterns are displayed in Figure 3.

6 Conclusion

This paper introduces a model in which credit constrained firms can choose between investment projects that differ in their duration, pledgeability of their returns, and liquidity risk. The model provides a novel theoretical underpinning for the countercyclicality of financing constraints based on the procyclicality of investment return pledgeability. We use this framework to explore how the consideration of firms' investment choice problem in the presence of financing frictions might affect the role of such frictions in amplifying or dampening shocks to the macroeconomy. A novel dampening mechanism of technology shocks is identified, which is based on time-varying financing constraints that affects firms' preference for the duration profile of their portfolio of investment projects. The dampening effect is shown to be quantitatively large, especially in the presence of general equilibrium interactions. On the other hand, this framework is able to account for the empirically documented cyclical variation in the composition of investment, a feature which most existing models studying the macroeconomic implications of financial constraints cannot account for.

One limit of the analysis is that it only considers the impact of unexpected shocks. Introducing aggregate uncertainty in the form of stochastic productivity or financing constraints processes could influence the results obtained in this paper, particularly if firms are able to write financial contracts based on the outcomes of these aggregate shocks. A framework with multiple aggregate shocks and two-dimensional firm heterogeneity becomes computationally demanding, however, and whereas a body of methodological research has provided feasible and accurate approximations for a subset of environments featuring one-dimensional heterogeneity in the household sector and aggregate uncertainty, similar work has yet to appear in more general contexts that could be applicable to an environment like the one described in this paper.

This paper highlights the importance for models designed to study the macroeconomic implications of credit constraints of considering the type of investment firms carry out, in addition to the amount. While this paper focuses on a particular set of dimensions of project heterogeneity, several other important dimensions remain to be analyzed in the context of a general equilibrium heterogeneous agent framework like the one presented in this paper, such as minimum investment size, degree of investment irreversibility, or volatility of returns. Models that incorporate these or other dimensions of heterogeneity could be fruitful avenues of research.

A Appendix: Numerical Solution

A.1 Steady State

In the steady state, equilibrium interest rate r_{ss} will be equal to the households' rate of time preference given that they are financially unconstrained, so

$$r_{ss} = \frac{1}{\beta} - 1.$$

Firms' optimization in the steady state is solved taking r_{ss} as given, and assuming aggregate productivity θ_t and financing frictions λ_t are constant at their steady state level. Value function iteration is used, and the firm's idiosyncratic state space is discretized. Along the dimension of asset holdings $a_{f,t}$ the grid contains 20 points starting at $a_{f,t} = 0$, and given that important non-linearities arise in the value function for low values of asset holdings, the grid's density decreases exponentially with asset holdings. Along the dimension of long-term capital holdings $k_{L,t}$ the value function is moderately concave for most of the region, due to the presence of convex costs of adjustment, and for this reason 40 equally-spaced gridpoints are chosen, again starting at $k_{L,t} = 0$.

The steady state distribution of firm asset and long-term capital holdings is calculated using the firms' optimal policies by performing 600 iterations of a firm sector composed of 250,000 firms. From the converged distribution we obtain the steady state aggregate output, aggregate dividends, aggregate net corporate borrowing and wages. In equilibrium, firms with large asset holdings save and those with low asset holdings borrow. It is the case for all combinations of parameter values considered that net borrowing of the firm sector is positive.

In the steady state equilibrium households' consumption must be equal to the sum of wages, dividends and the return on savings:

$$C_{ss} = r_{ss}A_{ss} + W_{ss} + D_{ss},$$

where aggregate savings A_{ss} are equal to aggregate net corporate borrowing.

A.2 Transitional Dynamics Following Temporary Shocks

The solution method to obtain the transitional dynamics proceeds as follows. The process is broadly similar for technology and financial shocks, and the few differences are pointed out.

1. We first establish the shock size and persistence

- 2. We guess the number of periods it will take for convergence back to the initial steady state. We use 150 periods for technology shocks, and 80 periods for financial shocks, which are less persistent.
- 3. We guess a path for r_t and for C_t for all the transition periods
- 4. We start backwards and, taking the t + 1 value function of firms as given, calculate optimal time t policies of firms and the resulting value function given (i) the path of the shock, and (ii) the conjectured path for $\{r_t\}$. We do this for every period, using the updated t + 1 value function when calculating the time t policies and value function.
- 5. Starting from the steady state distribution of firms, we simulate the entire firm sector (250,000 firms) for the duration of the transitional dynamics using the optimal policies calculated in the previous step. For each period, we calculate aggregate dividends D_t , aggregate net corporate borrowing (which we will equate to A_{t+1}) and wages W_t . Next, for each period we obtain the r_{t+1} that clears the bond market. We do so by first calculating the household consumption C_t^{eq} that would be consistent with a level of household savings equal to net corporate borrowing

$$C_t^{eq} = (1+r_t)A_t + W_t + D_t - A_{t+1}^{eq},$$

taking r_t from our guess and A_t from the previous period, and W_t , D_t and A_{t+1}^{eq} from the simulation of firms. Finally, we calculate the equilibrium interest rate that clears the bond market

$$r^{eq} = \frac{1}{\beta} \left(\frac{C_t^{eq}}{C_{t+1}} \right)^{-\gamma} - 1,$$

taking C_{t+1} as given from the conjectured path.

6. Update the conjectured path for r_t and C_t using a small updating parameter (0.02 in our case), and repeat 4-5 until convergence.

References

- Acharya, V.V., Almeida, H., Campello, M., 2007. Is cash negative debt? A hedging perspective on corporate financial policies. Journal of Financial Intermediation 16(4), 515-554.
- Aghion, P., Askenazy, P., Berman, N., Cette, G., Eymard, L., 2012. Credit constraints and the cyclicality of R&D investment: Evidence from France. Journal of the European Economic Association 10(5), 1001-1024.
- Aghion, P., Angeletos, G.-M., Banerjee, A., and Manova, K., 2010. Volatility and Growth: Credit Constraints and Productivity-Enhancing Investment. Journal of Monetary Economics 57, 246-65.
- Ahn, S., Denis, D.J., Denis, D.K., 2006. Leverage and investment in diversified firms. Journal of Financial Economics 79(2), 317-337.
- Almeida, H., Campello, M., Weisbach, M.S., 2011. Corporate financial and investment policies when future financing is not frictionless, Journal of Corporate Finance 17(3), 675-693.
- Almeida, H., Campello, M., Weisbach, M.S., 2004. The Cash Flow Sensitivity of Cash. Journal of Finance 59, 1777–1804.
- Andrade, G., Kaplan, S.N., 1998. How Costly is Financial (Not Economic) Distress? Evidence from Highly Leveraged Transactions that Became Distressed. Journal of Finance 53, 1443-1493.
- Angeletos, G.-M., Calvet, L.-E., 2006. Idiosyncratic production risk, growth and the business cycle. Journal of Monetary Economics 53, 1095-1115.
- Bancel, F., Mittoo, U.R., 2004. Cross-Country Determinants of Capital Structure Choice: A Survey of European Firms. Financial Management 33, 103-132.
- Bates, T.W., Kahle, K.M., Stulz, R.M., 2009. Why Do U.S. Firms Hold So Much More Cash than They Used To?. Journal of Finance 64, 1985-2021.
- Bernanke, B.S., Gertler, M., 1989. Agency Costs, Net Worth, and Business Fluctuations. American Economic Review 79, 14-31.
- Bernanke, B.S., Gertler, M., Gilchrist, S., 1999. The financial accelerator in a quantitative business cycle framework. Handbook of Macroeconomics, in: J. B. Taylor and M. Woodford (ed.) 1, 1341-1393.
- Bilbiie, F.O., Ghironi, F., Melitz, M.J., 2012. Endogenous Entry, Product Variety, and Business Cycles. Journal of Political Economy, 120(2), pages 304 - 345.
- Caggese, A., 2007. Financing constraints, irreversibility, and investment dynamics. Journal of Monetary Economics, 54 2102-2130.
- Caggese, A., Cuñat, V., 2008. Financing Constraints and Fixed-term Employment Contracts. Economic Journal 118, 2013-2046.
- Carlstrom, C.T., Fuerst, T.S., 1997. Agency Costs, Net Worth, and Business Fluctuations: A Computable General Equilibrium Analysis. American Economic Review 87, 893-910.
- Chevalier, J.A., Scharfstein, D.S., 1996. Capital-Market Imperfections and Countercyclical Markups: Theory and Evidence. American Economic Review 86, 703-25.

- Christiano, L., Motto, R., Rostagno, M., 2010. Financial factors in economic fluctuations. European Central Bank Working Paper.
- Cooper, R.W., Haltiwanger, J.C., 2006. On the Nature of Capital Adjustment Costs. Review of Economic Studies 73, 611-633.
- Dew-Becker, I., 2012. Investment and the Cost of Capital in the Cross-Section: The Term Spread Predicts the Duration of Investment. Federal Reserve Bank of San Francisco Working Paper
- Eisfeldt, A.L., Rampini, A.A., 2007. New or used? Investment with credit constraints. Journal of Monetary Economics 54, 2656-2681.
- Gertler, M., Kiyotaki, N., 2011. Financial Intermediation and Credit Policy in Business Cycle Analysis. Handbook of Monetary Economics, Vol. 3, ed. Benjamin Friedman and Michael Woodford.
- Gourio, F., Kashyap, A.K., 2007. Investment spikes: New facts and a general equilibrium exploration. Journal of Monetary Economics 54, 1-22.
- Graham, J., Campbell, H., 2001. The theory and practice of corporate finance: Evidence from the field. Journal of Financial Economics 60, 187–243.
- Hall, R. E., 2011. The High Sensitivity of Economic Activity to Financial Frictions. The Economic Journal 121, 351–378.
- Hart, O., Moore, J., 1994. A Theory of Debt Based on the Inalienability of Human Capital. Quarterly Journal of Economics, MIT Press, vol. 109(4), pages 841-79, November.
- Holmstrom, B., Tirole, J., 1997. Financial Intermediation, Loanable Funds, and the Real Sector. Quarterly Journal of Economics 112, 663-91.
- Iacoviello, M., 2005. House Prices, Borrowing Constraints, and Monetary Policy in the Business Cycle. American Economic Review 95, 739-764.
- Jensen, M.C., 1986. Agency Costs of Free Cash Flow, Corporate Finance, and Takeovers. American Economic Review 76, 323-29.
- Khan, A., Thomas, J.K., 2011. Credit Shocks and Aggregate Fluctuations in an Economy with Production Heterogeneity. NBER Working Paper 17311.
- Kiyotaki, N., Moore, J., 1997. Credit Cycles. Journal of Political Economy 105, 211-248.
- Matsuyama, K., 2007. Credit Traps and Credit Cycles. American Economic Review 97, 503-516.
- Peyer, U.C., Shivdasani, A., 2001. Leverage and internal capital markets: evidence from leveraged recapitalizations. Journal of Financial Economics 59, 477-515.
- Smets, F., Wouters, R., 2007. Shocks and Frictions in US Business Cycles: A Bayesian DSGE Approach. American Economic Review 97, 586-606.
- Weston, J.F., Siu, J.A., 2003. Changing Motives for Share Repurchases. UCLA Working Paper.